Author's response to reviews

Title: Preliminary study of the antioxidant properties of the flowers and roots of Pyrostegia venusta (Ker Gawl) Miers

Authors:

Purabi Roy Mrs (purviroy@gmail.com)
Sarika Amdekar Ms. (ananyasarika@gmail.com)
Avnish Kumar Dr. (avnishkumar81@gmail.com)
Vinod Singh Dr. (vsinghimuunology@gmail.com)

Version: 5 Date: 11 August 2011

Author's response to reviews: see over
Dear Sir or Madam:

We are grateful for giving us an opportunity to revise our manuscript. Enclosed please find the revised manuscript entitled “Preliminary study of antioxidant property of flowers and roots of Pyrostegia venusta (Ker Gawl) Miers.” The initial manuscript was copy edited as advised. We have addressed the query of the respected reviewer. The manuscript was again read by a professional editing service to improve the quality of English. The manuscript was also read and revised by all authors to improve the overall quality.

We would highly appreciate if you kindly consider the paper favorably.

Thanking you,

Sincerely yours,

Dr. Vinod Singh